Author's response to reviews

Title: Risk factors for antenatal vitamin D insufficiency in an urban district in Malaysia

Authors:

Norikliil Bukhary Ismail Bukhary (norikliil@yahoo.com)
Zaleha Md Isa (zms@ppukm.ukm.edu.my)
Khadijah Shamsuddin (khadijah@ppukm.ukm.edu.my)
Geok Lin Khor (geoklin_khor@imu.edu.my)
Zaleha Abdullah Mahdy (zaleha@ppukm.ukm.edu.my)
Haslinda Hassan (lynn_nidz@yahoo.com.my)
Noor Sharifatul Hana Yeop (sharifatul_hana@yahoo.com)

Version: 2
Date: 18 March 2015

Author's response to reviews:

Below are the changes:
1) Include line and page numbers in the manuscript
2) Remove page break in the manuscript